Author’s response to reviews

Title: The implementation of the Medical Regulation Office and Mobile Emergency Attendance System and its impact on the gravity profile of non-traumatic afflictions treated in a University Hospital: a research study

 Authors:
 Sergio L B Lopes (siblopes@uol.com.br)
 Jose Sebastiao DOS Santos (jsdsanto@fmrp.usp.br)
 Sandro Scarpelini (sandro@fmrp.usp.br)

Version: 5 Date: 12 May 2007

Author’s response to reviews:

THE AUTHORS DEMAND 07 MORE DAYS TO ANSWER INTEGRALLY ONE OF THE REVIEWERS. SO, IF POSSIBLE, DO NOT CONSIDER THE PREVIOUS ANSWER (REVISIONS PROVIDED-2) AND GIVE US SOME MORE DAYS. THIS FACT IS BASED ON A NEW STATISTIC HELP PROVIDED BY OUR UNIVERSITY.

THANKS.